Author's response to reviews

Title: p53 expression is significantly correlated with high risk of malignancy and epithelioid differentiation in GISTs. An immunohistochemical study with 104 GISTs.

Authors:

Ursula Pauser (upaus@gmx.de)
Christian Kaehler (kaehler@patho.uni-luebeck.de)
Günter Kloeppel (gkloeppel@path.uni-kiel.de)
Hartmut Merz (merz@patho.uni-luebeck.de)
Alfred C Feller (feller@patho.uni-luebeck.de)

Version: 2 Date: 28 March 2008

Author's response to reviews:

I completed the requests according the editorial board listed:
1. Title page, Abstract and Authors are listed in the manuscript
2. There are no competing interests
3. Authors` contribution is listed
4. The retrospective study carried out in compliance with the Helsinki Declaration